Author's response to reviews

Title: Managing co-morbid depression and anxiety in primary care patients with asthma and/or Chronic Obstructive Pulmonary Disease: design of a randomized controlled trial.

Authors:

Antoinette M. Pommer (A.M.Pommer@uvt.nl)
François Pouwer (F.Pouwer@uvt.nl)
Johan Denollet (J.Denollet@uvt.nl)
Victor JM Pop (V.J.M.Pop@uvt.nl)

Version: 3 Date: 2 January 2012

Author's response to reviews: see over
To the editors:
Prof. dr. Altman
Prof. dr. Furberg
Prof. dr. Rothwell
Prof. dr. Grimshaw
BCM Trials

Tilburg January, 2012

Dear Professor Altman, dear Professor Furberg, dear Professor Rothwell, dear Professor Grimshaw,

Please find enclosed a revised version of the manuscript:

‘Managing co-morbid depression and anxiety in primary care patients with asthma and/or Chronic Obstructive Pulmonary Disease: design of a randomized controlled trial’

All proposed changes to the format of the manuscript so that it conforms to the journals style have been adopted.

1. Title: please ensure the title conforms to journal style for study protocol articles. The title should follow the format ‘__________: study protocol for a randomized controlled trial’

2. Can you please also ensure that your study protocol is formatted as outlined in our instructions for authors. Please make sure to include all sections listed and to order them correctly. Please note that figures and tables should not be included in the main body of the manuscript.

sincerely,

Antoinette Pommer, MSc

Tilburg University
P.O. Box: 90153
5000 LE Tilburg
The Netherlands